Author's response to reviews

Title: Program of active aging in a rural Mexican community: a qualitative approach

Authors:

Maria de la Luz Martinez-Maldonado (marilumtz05@yahoo.com.mx)
Elsa Correa-Munoz (elcomm@terra.com.mx)
Victor Manuel Mendoza-Nunez (mendovic@servidor.unam.mx)

Version: 2 Date: 13 July 2007

Author's response to reviews: see over
Dear Dr. Phillips:

I send you the corrected manuscript entitled “Program of active aging in a rural Mexican community: a qualitative approach” and reply point-by-point to comments of the reviewers, for review and possible publication in BMC-Public Health.

Sincerely

Víctor Manuel Mendoza-Núñez, MD, PhD (Corresponding author)
FES Zaragoza, UNAM
Batalla 5 de mayo s/n
Esq. Fuerte de Loreto
Col. Ejército de Oriente
Mexico City 09230
México

E-mail: mendovic@servidor.unam.mx
1. Comment:

The Gerontological care model (Figure 1) needs to be revised to account for the relationship between the Gerontological health care unit and training of the gerontological promoters. From the description in the text, the Unit guides the Training, but this is not represented in the Figure. The other point that needs clarification is the use of the terms "control" for healthy and ill elderly care needs, and "enforcement" of self-care that is described in the Figure legend. These terms seem counterintuitive in a model that purportedly supports empowerment and self-care. From the text description, it's also unclear why there are 2 GN for each of the 4 program objectives in the model, and how the GN contribute to the desired outcomes.

1. Reply:

It was corrected the figure and changed the terms.

2. Comment:

The study intervention consisted of a 14-week educational program. The program description is too limited, and should include more detail about what information was provided, and the circumstances of the educational sessions. There were 155 participants, but how the program was provided for these participants is not identified.

2. Reply:

This information was included.

3. Comment:
Overall, the methods and analytic strategies used in this project require more clarification. For example, one of the key findings is that participants began to "organize among them and start integrating groups", but no supporting evidence is provided to justify or explain this finding. This same issue is encountered throughout the methods and results sections. The conclusions require more supporting data and explanation.

3. Reply:

The information requested was included in methods, results and discussion sections.

4. Comment:

The authors indicate that a "qualitative analysis" of various data sources led to specific conclusions (pg. 10). These explanations and links between the data and the interpretation need to be made clear.

4. Reply:

This information was included.

5. Comment:

The report should include other available data related to descriptions of the participants, such as, participation rates in the educational sessions, retention rates (100 out of 155), explanation of drop-out, what is meant by "functioning as paraprofessional gerontologists", and the outreach activities to neighboring communities following completion of the program. It's unclear what the process was for the 100 program completers to extend their sphere of influence to 500 older adults in other communities.

5. Reply:

This information was included.
6. Comment.

“Several grammatical errors need to be corrected. For example, in the abstract section for methods (last sentence, i.e., ..for measure the) and conclusion (first sentence, i.e., ...for reach the), "for" is used instead of the correct "to".

6. Reply:

The manuscript was edited.

7. Comment:

In the Background section, the authors use the phrase 4 key elements, yet the elements appear to be beliefs of the authors, so it would be more correct to identify these as beliefs that provide the framework for the study. These beliefs must be linked to and explained in the context of the Gerontological care model (Figure 1) that is presented as a model for the study.

7. Reply:

These aspects were corrected.